Author’s response to reviews

Title: Magnesium Deficiency: Effect on Bone Mineral Density in the Mouse Appendicular Skeleton

Authors:

Dr Helen E. Gruber (hgruber@carolinas.org)
Robert K. Rude (RRude60075@aol.com)
Livvy Y. Wei (livvywei@yahoo.com)
Angie Frausto (frausto@usc.edu)
Barbara G. Mills (bmills@hsc.usc.edu)
H. JAMES Norton (jnorton@carolinas.org)

Version: 4 Date: 15 Apr 2003

Dr. Singer had no more corrections in his review. Dr. Howard identified a typo which has been corrected, and suggested addition of statistical comments in the legend of table 1. This has also been done. The order of items at the end of the manuscript has been corrected as requested by the editor. As requested also by the editor, I have had our AV department scan the figures and new files will be submitted now. This should complete all requested changes. Thank you for the opportunity to revise this manuscript.